Author's response to reviews

Title: Early intervention program reduces stress in parents of pre-terms during childhood; a randomized controlled trial.

Authors:

Inger P Landsem (inger.pauline.landsem@unn.no)
Bjørn H Handegård (bjorn.helge.handegaard@iut.no)
Jorunn JT Tunby (jorunn.tunby@uit.no)
Stein E Ulvund (s.e.ulvund@ped.uio.no)
John A Rønning (john.ronning@uit.no)

Version: 5 Date: 2 July 2014

Author's response to reviews:

The letter uploaded in June 27 is the last. This re-submission is answering the comments from the editorial services received 01.July.